Author's response to reviews

Title: How well do second-year students learn physical diagnosis? Observational study of an objective structured clinical examination (OSCE)

Authors:

Claus Hamann (chamann@partners.org)
Kevin Volkan (kevin.volkan@csuci.edu)
Mary B Fishman (MBF3@gunet.georgetown.edu)
Ronald C Silvestri (rsilvest@caregroup.harvard.edu)
Steven R Simon (steven_simon@hms.harvard.edu)
Suzanne W Fletcher (suzanne_fletcher@hms.harvard.edu)

Version: 3 Date: 14 Dec 2001

Dear Editor,
Thank you again for your acceptance of our article. The reviewer of our Nov. 22 re-submission indicated no further changes were required. Our final manuscript includes changes requested on the checklist: citation and description of Additional Files in the body and at the end of the manuscript, and numbering format for the references. Table 1 now lists the OSCE stations alphabetically.
Sincerely,
Claus Hamann MD,MS